Math 711, Fall 2004

Instructor: Mel Hochster Office: 3078 East Hall e-mail: hochster@umich.edu

Hard copies of lectures will be distributed, usually at the next lecture. The PostScript files will also be distibuted by e-mail and will be posted on my Web page.

Grades will be based on problem sets: these will be distributed, roughly, every other week. There will be five problem sets, with approximately 25–30 problems in total. There will not be any exams.

Tentative office hours:

MWF 3–4 p.m.

I will also be available, typically, for 20 minutes before class and for 10 minutes after class. I am usually quick to answer questions by e-mail. Of course, you may also make separate appointments.